Author’s response to reviews

Title: Polypharmacy in a hospitalized psychiatric population: risk estimation and damage quantification

Authors:

JAIME CARMONA-HUERTA (JAIME.CARMONA@ACADEMICOS.UDG.MX)
SANTIAGO CASTIELLO-DE-OBESO (santiagocdo@gmail.com)
JULISSA RAMIREZ-PALOMINO (julissa.areli.palomino@hotmail.com)
RODRIGO DURAN-GUTIERREZ (med04rodrigo@gmail.com)
DAVID CARDONA-MULLER (david.cardona@academicos.udg.mx)
FERNANDO GROVER-PAEZ (fgroverp@hotmail.com)
PAOLA FERNANDEZ-DORANTES (pa-fdezd@hotmail.com)
RAFAEL MEDINA-DAVALOS (medina_666@hotmail.com)

Version: 4 Date: 06 Feb 2019

Author’s response to reviews:

Corrections:

• Page 1 Section “Abstract”
  o The title “Introduction” was changed for “Background”

• Page 3 Section “Manuscript body”
  o The title “Introduction” was changed for “Background”

• Page 4
  o The title “Design and methods” was changed to “Methods”

• Page 25 Section “Declarations”
  o The title “Consent to publish” together with the legend “not applicable” were added
• Page 23 Section “Declarations”
  o Declarations section moves before References section
* Page 23 Section “Authors’ contributions”
  o The name of the authors were organized as specified (full initials, not full name)
  o The subtitle “Conflict of interest” was changed to “Competing interests”
  o The subtitle “Role of the funding source” was changed to “Funding”
• Page 27
  o The section “Declaration of author agreement” was removed from the final manuscript.